PLOS NEGLECTED TROPICAL DISEASES



G OPEN ACCESS

Citation: Dana D, Vlaminck J, Ayana M, Tadege B, Mekonnen Z, Geldhof P, et al. (2020) Evaluation of copromicroscopy and serology to measure the exposure to *Ascaris* infections across age groups and to assess the impact of 3 years of biannual mass drug administration in Jimma Town, Ethiopia. PLoS Negl Trop Dis 14(4): e0008037. https://doi.org/10.1371/journal.pntd.0008037

Editor: John Stuart Gilleard, University of Calgary, CANADA

Received: May 17, 2019

Accepted: January 8, 2020

Published: April 13, 2020

Copyright: © 2020 Dana et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: This study was funded by the VLIR-UOS program Belgium with Jimma University, Ethiopia and a Special Research Fund of Ghent University, Belgium. Both JV and BL received funds from the Research Foundation (www.FWO.be). The funders had no role in study design, data collection and

RESEARCH ARTICLE

Evaluation of copromicroscopy and serology to measure the exposure to *Ascaris* infections across age groups and to assess the impact of 3 years of biannual mass drug administration in Jimma Town, Ethiopia

Daniel Dana^{1,2}, Johnny Vlaminck^{2*}, Mio Ayana¹, Bamlaku Tadege^{2,3}, Zeleke Mekonnen¹, Peter Geldhof², Bruno Levecke^{2*}

 School of Laboratory Science, Faculty of Health Science, Institute of Health, Jimma University, Jimma, Ethiopia, 2 Department of Virology, Parasitology and Immunology, Ghent University, Merelbeke, Belgium,
 School of Medicine, Hawassa University, Hawassa, Ethiopia

* Johnny.vlaminck@ugent.be (JV); Bruno.levecke@ugent.be (BL)

Abstract

Background

The scientific community has recently summarized the desired characteristics for diagnostic tools across the different phases of a soil-transmitted helminth (STH) mass drug administration (MDA) program. Although serology meets some of the desired criteria, there is a scarcity of data on baseline serological profiles in human populations, both prior to and during MDA programs.

Methods

In this study, we compared the copromicroscopic and the serological infection profiles in 600 school-aged children (SAC) and 600 adults at the advent of the MDA program in Jimma Town, Ethiopia. The serological profiles were examined by two ELISAs that measure IgG4 responses to the *Ascaris suum* haemoglobin antigen (AsHb) and a somatic extract of lung stage larvae (AsLungL3). Three years into the MDA program, we sampled another group of 600 SAC from the same schools to assess the reduction in prevalence and intensity of *Ascaris infections* measured by copromicroscopy and serology.

Principal findings

Prior to the start of MDA, copromicroscopy revealed an *Ascaris* prevalence of 31.0% and a mean fecal egg count of 2,919 eggs per gram (EPG) in SAC. Following three years of biannual treatment, the prevalence reduced to 13.2% (57.8% reduction) and the mean fecal egg count to 1,513 EPG (48.1% reduction). This reduction was also reflected in the serological results. The seroprevalence reduced with 40.9% and 27.4% and the mean optical density ratio reduced with 44.2% and 38.2% as measured by the AsHb or AsLungL3 ELISA

analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

respectively. We also showed that, despite a decreasing coproprevalence, seroprevalence to *Ascaris* increased with age.

Conclusions

This study is the first to provide IgG4 response profiles of an endemic population to two different *A. suum* antigens. The results suggest that exposure to the infectious stages of *Ascaris* reaches beyond SAC alone. Furthermore, it highlights the possible use of serological assays to monitor changes in STH exposure during MDA programs.

Author summary

Worldwide, there has been an upscale in deworming programs to reduce the disease burden attributable to intestinal worms (giant roundworms, whipworms and hookworms). As a consequence of this, there is a focus-shift from morbidity control towards breaking transmission and ultimately disease elimination. Current standard diagnostic tools, which are based on the demonstration and quantification of worm eggs in stool (copromicroscopy), lack the diagnostic performance to make well-founded decisions when a program reaches its endgame. The scientific community has therefore defined minimal criteria that new potential diagnostic methods should meet. Diagnostic tools that are based on the detection of antibodies, which are indicative of the natural immune response of the host to worm-specific components in blood, potentially meet these criteria. We compared giant roundworm infection profiles by copromicroscopy and two antibody-based tests in both children and adults in an endemic setting. Three years into the national deworming program, another group of children from the same schools were sampled to evaluate the program progress. Our results provide a first step in the evaluation of antibodies as a diagnostic tool in the endgame of deworming programs, but they also highlight the need for further research on the interpretation of antibody-based diagnostics for STH.

Introduction

In 2012, more than 70 pharmaceutical companies, governments, and global health organizations signed the London Declaration on Neglected Tropical Diseases (NTDs) and committed themselves to control and eliminate NTDs at the individual patient level or at the population level through programs of mass drug administration (MDA). Soil-transmitted helminthiasis is caused by a group of intestinal worms (*Ascaris lumbricoides, Trichuris trichiura* and two hookworm species (*Ancylostoma duodenale* and *Necator americanus*)) and is considered one of the most important NTDs that is amenable to control through MDA. It has been estimated that these worms affect 1.45 billion people, which accounts for 11% of the global burden attributable to NTDs [1, 2]. So far, major progress has been made in combating soil-transmitted helminthiasis, significantly reducing its global disease burden [3]. This, together with the prospect of sustained anthelmintic drug donations, has caused a focus-shift from morbidity control towards breaking transmission and ultimately disease elimination [4, 5].

Disease elimination programs go through a variety of phases, and programmatic decisions to transition from one program phase to the next are driven by the outcome of large epidemiological surveys applying standard diagnostic tools [6–8]. Currently, the microscopic detection and quantification of eggs excreted by adult worms in stool remains the standard diagnostic tool [9]. However, copromicroscopy lacks the performance to make well-founded decisions when a program reaches its endgame [10, 11]. In fact, implementing these methods may potentially lead to substantial health and monetary consequences resulting from a premature stop decision.

As a response to this urgent need for improved diagnostic tools, the community summarized the desired characteristics-the so called target product profiles—for the different phases of the program (use-case #1: determine disease transmission and identify type of MDA, usecase #2: assess progress against program goals, use-case #3: confirm a decision to stop intervention and transition into surveillance and use-case #4: verify a sustained break in transmission; [10]). For the last two use-cases, the community concluded that the desired diagnostic biomarker should be in sufficient abundance in a readily accessible body fluid, excluding stool. Potential biomarkers that meet this desired characteristic are infection-induced host antibodies. Measuring the presence and concentration of these antibodies, as a proxy for parasite exposure, has already been proven helpful in the elimination program of lymphatic filariasis, another MDA amenable NTD [12].

Recent studies have proven the usefulness of serological assays for *Ascaris suum* diagnosis in pigs. Serum antibody responses against an antigen from adult worms (*A. suum* haemoglobin (AsHb) and the total extract of *A. suum* lung stage L3 larvae (AsLungL3) showed higher sensitivity than copromicroscopy to identify exposure to both experimental and natural *A. suum* infections [13, 14] and provided more accurate assessments of disease burden [14–18]. It is however difficult, if not impossible, to obtain similar high-quality data from studies in human populations. Nevertheless, a recent study showed a significant reduction in serum antibody response to AsHb in an Indonesian human population during the course of a community-based MDA control program targeting lymphatic filariasis and soil-transmitted helminthiasis, already highlighting the potential of serology to assess progress against program goals in STH control programs (use case #2) [19].However, today, there is a scarcity of data on baseline serological profiles to these antigens in human populations of endemic and non-endemic areas. This limits our insights into biomarker-specific infection profiles and impedes the evaluation of the clinical utility of serology in the different phases of a MDA program.

Therefore, the main aims of this study were (i) to assess diagnostic cutoffs for both AsHb and AsLungL3 using sera from a non-endemic population, (ii) to compare the serological profiles of school-aged children (SAC) and adults from an endemic population in Ethiopia with infection profiles obtained by copromicroscopy and (iii) to evaluate the impact of three years of bi-annual MDA on STH infections applying copromicroscopy and *Ascaris* serology as a proof of concept.

Methods

Ethical considerations

Blood samples from the non-endemic adult population in Belgium were acquired through Red Cross Flanders (order numbers: CG2016 0404A, CM2016 0627B and CG2016 1219F). For the collection of stool and blood samples in Jimma, Ethiopia, ethical approval was obtained from the Institutional Review Board of both Ghent University, Belgium (reference number: 2015/0801 and study registration number: B670201526293) and Jimma University, Ethiopia (reference number: RPGC/181 and IHRPGD/680). For SAC, the school authorities, teachers, parents, and the children were informed about the purpose and procedures of the study. A written informed consent was obtained from the parents/guardians. An additional separate written consent was secured from children older than 12 years. Consent forms were prepared in English, translated into the two commonly used local languages (Afaan Oromo and

Amharic) and a copy was handed over to the children's parents/guardians. Only those children who were willing to participate and whose parents or guardians signed the written informed consent form were included in the study. For the adult population, we only involved subjects in the study that provided their signed informed consent before data collection. All individuals infected with any STH based on copromicroscopy were administered a single oral dose of 400 mg albendazole (GlaxoSmithKline).

Study design

Three cross-sectional surveys were performed in Jimma Town, Southwest Ethiopia. In a first survey (October–December 2015), we screened 600 SAC from ten primary schools across 6 *kebeles* (local language for neighborhoods). The selection of the primary schools was based on the presence of the age groups of interest (5 to 18 years) and their participation in previous studies [20–22]. In each school, 60 subjects were sampled (30 children between the age 5 and 10, and 30 children between the age of 14 and 18), resulting in a total of 600 SAC. In a second survey (February to May 2017), we screened 600 adults across the 6 *kebeles* (100 per *kebele*). After three years of bi-annual school-based MDA, a third survey was conducted from November to December 2018. This survey included another sample of 600 randomly selected SAC from the same ten primary schools that were included during the first survey. An overview of the number of SAC and adults sampled per *kebele* is provided in S1 Table.

Individuals consenting to participate in the study were asked to provide at least 5 g of fresh stool in a clean and labeled stool container. Because we wanted to maximize the clinical sensitivity of our analysis and produce the most accurate infection profiles (minimizing the number of false negative cases) in survey 1 (SAC) and 2 (adults), we processed all stool samples by Kato-Katz (0.0417 g of stool), McMaster egg counting method (2 g of stool) and Mini-FLO-TAC (2 g of stool) for the detection and enumeration of STH eggs. Each of these methods has been validated for diagnosis of STH infections [23]. In the third survey (SAC), stool samples were only examined by Kato-Katz, as this is the current recommended diagnostic method to evaluated progress made in MDA programs for STH. In addition to the stool sample, blood was collected from each individual. Blood samples were centrifuged for 10 min at 4,000 g and 4°C. Subsequently, the serum was recovered and stored at -20°C. All sera were shipped on dry ice to the Laboratory of Parasitology at Ghent University (Belgium) for serological analysis.

Copromicroscopy

Kato-Katz. Kato-Katz thick smears (Kato-Katz) were prepared per individual as described by WHO [9, 24]. To avoid clearance of hookworm eggs, all smears were examined within 30–60 min for the presence of STH eggs. The number of STH eggs was counted as per species basis and multiplied by 24 to obtain the fecal egg counts (FECs; expressed in eggs per gram of stool (EPG)).

McMaster. The McMaster egg counting method was performed as previously described [25]. In short, 2 g of stool was suspended in 30 ml of flotation solution (saturated salt solution at room temperature with specific density ~1.20). The stool suspension was poured three times through a wire mesh to remove large debris. Then, aliquots of 0.5 ml were transferred to each of the two chambers of a McMaster slide. Finally, the numbers of STH eggs were counted in each of the two chambers using light microscopy at a 100x magnification. The FECs were obtained by multiplying the total number of eggs by 50 for each of the STH species.

Mini-FLOTAC. The Mini-FLOTAC was performed as described by Cringoli and colleagues [24, 26, 27]. In short, 2 g of fresh stool was weighed into the Fill-FLOTAC container, followed by addition of 38 ml of flotation solution (same as for McMaster). Then the stool was

thoroughly homogenized using the homogenizer stick of the Fill-FLOTAC. The fecal suspension was filtered through the Fill-FLOTAC and subsequently added to both chambers of the Mini-FLOTAC device. After 10 minutes, the top part of flotation chambers was translated and the numbers of STH eggs were counted in each of the two chambers using light microscope at a 100x magnification. The FECs were obtained by multiplying the total number of eggs by 10 for each of the STH species.

Serology. Serological analysis of the serum samples was performed based on the protocols described by Vlaminck *et al.* for the AsHb ELISA [19] and Vandekerckhove *et al.* for the AsLungL3 ELISA [13]. These ELISAs are designed to measure IgG4 responses as it has been shown by Santra et al., 2001 [28] and Chatterjee et al., 1996 [29] that human IgG4 responses to *Ascaris* antigens showed less cross reactivity than IgG1, IgG2 and IgG3 subclass antibodies in sera from patients infected with hookworm, *Trichuris* and *Strongyloides*.

Maxisorp 96-well ELISA plates were coated overnight at 4°C with 1µg/ml AsHb and 5µg/ ml of AsL3Lung in carbonate buffer (pH = 9.6). After three wash steps with the wash buffer (PBST: phosphate buffered saline containing 0.05% Tween20), non-specific binding sites were blocked for 1 hour at room temperature using blocking buffer (PBS + 5% heat treated fetal calf serum). The coated antigens were detected by adding diluted sera and controls (positive-(PC) and negative controls (NC)) at a 1:200 dilution for AsHb and 1:400 dilution for AsLungL3 in PBST for 2 hours at room temperature. Positive and negative controls were added to each plate. All samples were run in duplicate. Following three subsequent wash steps, horseradish peroxidase conjugated anti-human IgG4 (1: 20,000 dilution) in blocking buffer was added to the plate for 1 hour at room temperature to detect IgG4 antibodies that were bound to the antigens. After a final three washing steps with PBST, 100 µl of substrate was added (O-phenylenediamine 0.1% in citrate buffer (pH = 5.0)). The development reaction was stopped after 10 min with $2.5M H_2SO_4$ and the optical density (OD) measured at 492 nm. Reactivity to the AsHb and AsLungL3 antigen is shown in optical density ratio (ODr = $(OD_{sample}-OD_{NC})/(OD_{PC}-OD_{NC}))$. A collection of 500 blood samples from healthy Belgian blood donors were also screened to determine the diagnostic cutoff (positive vs. negative test result). The cutoff was established as mean ODr \pm 3x standard deviation.

To exclude potential cross-reactivity with *Toxocara spp*, a subset of blood samples from healthy Belgian donors was evaluated for antibodies against *Toxocara* L3 excretory/secretory proteins using a commercial IgG ELISA kit (Cypress diagnostics, Belgium). This ELISA was performed according to the manufacturer's instructions. Samples were classified as positive (ODr >1.1), negative (ODr <0.9) or equivocal $(1.1 \ge ODr \ge 0.9)$.

Statistical analysis. For surveys 1 and 2, the detection of patent STH infections (coproprevalence) was based on the combined test result of single Kato-Katz, McMaster and Mini-FLOTAC. An individual was considered to be infected with a particular STH species, when at least one egg was found in his/her stool sample using any of the three copro-microscopic methods. The intensity of the infections was estimated by taking the average FEC over the three copro-microscopic methods (surveys 1 and 2). The detection of exposure to *Ascaris* infection as measured by the IgG4 antibody response to parasite antigens (AsHb and AsLungL3) was determined as the percentage of individuals that tested positive on the serological assay (test-ODr > diagnostic cutoff) (seroprevalence). To be able to calculate the reduction in mean ODr measured by both assays that can be compared to the reduction in mean FECs, all negative ODr values (i.e. samples with OD values below that of the negative reference sample) were given a zero value.

Statistically significant differences in qualitative (positive and negative test result) *Ascaris* test results were assessed using the 'glmer' function in R. The test-result (positive or negative) was used as dependent variable while the parameters: 'test' (3 levels: copromicroscopy,

AsLungL3 and AsHb), 'kebele' (9 levels: Awetu, Mendera, Bacho Bore, Bossa Addis, Bossa Kitto, Ginjo, Hermata, Jiren, Mentina, Seto Semaro), 'sex' (2 levels: male and female) and 'age of subject' (in years) were included as potential predicting variables. For a simplified interpretation of the effect of age, the age was centralized around the median age (18 years).

In addition, we assessed the reduction in both prevalence and intensity of STH infections. For copromicroscopy, the infection intensity was measured as the mean FECs and the proportion of low, moderate-to-heavy intensity infections as defined by WHO [30]. For serology, the intensity of exposure was measured as the mean ODr obtained by ELISA. The 95% confidence intervals (95% CI) around the point estimates were determined through bootstrap analysis (5,000 iterations) for all infection parameters. We concluded that reduction in prevalence or intensity was significant when the 95% CI interval did not include zero. Finally, correlations between ODr and mean FEC were determined using the Pearson correlation coefficient. The level of significance was set at p < 0.05.

Results

Study population

A total of 1,800 participants were sampled during the 3 surveys performed in this study. Namely, 1,200 SAC from 10 primary schools (600 in 2015 and 600 in 2018) and 600 adults from the neighborhoods (*kebeles*) where the schools were located. All of the participants provided sufficient stool to perform all copro-microscopic methods and a sufficient amount of blood to perform serology.

STH infection profiles in the population prior to MDA

The cross-sectional surveys of SAC and adults at the start of the MDA program resulted in a total study population of 704 female and 496 male individuals across 9 *kebeles* (Awetu Mendera: 60; Bacho Bore: 160; Bossa Addis: 160; Bossa Kitto: 160; Ginjo: 220; Hermata: 60; Jiren: 60; Mentina: 160; Seto Semaro: 160) and 5 age groups (age 5 to 10: 300; age 14 to 17: 263; age 18 to 29: 310; age 30 to 49: 208; at least 50 years of age: 119).

The goal of the first two surveys was to get an overview of the spread and intensity of STH infections in the whole community in Jimma Town, Ethiopia. To increase diagnostic sensitivity, stool samples were analyzed using three copro-microscopic methods (Kato-Katz, McMaster and Mini-FLOTAC) and the combined result of these analysis was used to determine the prevalence of any STH. The overall presence of STH in the whole population (1,200 individuals) was 52.9% (635/1,200). Trichuris infections were the most prevalent (40.4%), followed by Ascaris (22.3%) and hookworm infections (13.8%). Males were more infected than females for each of the three STH species (Ascaris: 27.2% vs. 18.9%; Trichuris: 44.8% vs. 37.4%; hookworms: 16.7% vs. 11.8%). The prevalence for each STH per age group is illustrated in Fig 1. This revealed that the highest prevalence for Ascaris (39.7%) occurred in the youngest age group (age 5 to 10). Ascaris prevalence steadily dropped as individuals grew older, with only 3.4% of individuals older than 50 excreting Ascaris eggs. Across the different age groups, Trichuris infection was most common in teenagers of 14 to 17 years of age (53.6%). Trichuris prevalence also reduced with age, but it never dropped below 25%. Hookworms were the least prevalent of all STHs, with a maximum prevalence of 22.4% in SAC aged 14 to 17. Just like for the other two STH species, hookworm prevalence dropped in the adult population from 18.1% in the 18 to 29-year-olds to 5.3% in the 30 to 49-year-olds and 6.7% in the individuals of age 50 and above. Generally, the mean FEC across the five age groups and both sexes followed trends similar to that of prevalence, with the highest mean FECs observed in the age groups/sex with





the highest prevalence. **S1 Table** provides the prevalence and intensity of *Ascaris*, *Trichuris* and hookworm infections across the five age groups, both sexes and nine *kebeles*.

STH infection profiles in SAC after 3 years of bi-annual MDA

The same 10 primary schools that participated in 2015 (survey 1) were re-sampled in 2018 (survey 3) to assess the impact of three years of bi-annual MDA on the prevalence and intensity of STH infections. Stool samples collected from the 600 SAC in 2018 were only evaluated by Kato-Katz. This is because WHO recommends single Kato-Katz for monitoring the progress of STH MDA programs [30]. To compare the infection outcomes between both surveys, only Kato-Katz results were used. The copromicroscopic results of for both surveys are provided in Table 1.

In 2015, any STH infections were detected in 57.7% of sampled children, with Trichuris being the most prevalent (39.5%) followed by Ascaris (31.3%) and hookworm infections (10.3%). The prevalence of moderate-to-heavy intensity was 12.3% for any STH infections, 11.3% for Ascaris and 2.5% for Trichuris. All hookworm infections were of low intensity. The mean FEC followed a similar pattern across the STH species, the mean FEC being highest for Ascaris (2,919 EPG) and the lowest for hookworm infections (23 EPG). For Trichuris, mean FEC equaled 165 EPG. In 2018, following three years of biannual deworming, the prevalence dropped to 28.0% for any STH infection, 18.2% for Trichuris, 13.3% for Ascaris and 5.0% for hookworm infections. This corresponds to a significant (the 95% confidence interval does not zero) reduction in prevalence of 51.5% (95CI: 44.4-58.2), 57.5% (95CI: 46.8-66.7), 53.9% (95CI: 44.7-62.4) and 51.5% (95CI: 28.8-69.1), respectively. In line with these reductions in prevalence, there was also a reduction in infection intensity (measured as the reduction in prevalence of low and moderate-to-heavy intensity infections and mean FECs) for all STH species. The reduction for these parameters varied ranged from 48.0% to 67.3%, and was significant for all parameters except for Trichuris. For this STH, a non-significant reduction in moderate-to-heavy intensity infections was observed (48.0%, 95CI: -25.0-63.3). No moderateto-heavy intensity hookworm infections were detected in 2018.

Table 1. The impact of deworming on soil-transmitted helminth (STH) infections in school-aged children (SAC) measured by copromicroscopy and serology.Copromicroscopic data is expressed as the prevalence with 95% confidence interval (95 CI) of any, low or moderate-to-high (M-H) intensity infections. The mean fecalegg count (FEC) is expressed in eggs per gram (EPG) stool. Serological response to Ascaris suum haemoglobin (AsHb) or total lung stage L3 extract (AsLungL3) is reportedas mean Optical Density ratio (ODr) as measured by ELISA.

		2015 (n = 600)	2018 (n = 600)	Reduction (%) (95% CI)
Corpromicroscropy				
Any STH	Prevalence (%) (95 CI)			
	Any intensity	57.7 (53.8; 61.7)	28.0 (24.5; 31.5)	51.5 (44.4; 58.2)
	Low intensity	45.4 (41.3; 49.3)	22.2 (18.8; 25.5)	51.1 (42.5; 59.3)
	M-H intensity	12.3 (9.8; 15.2)	5.8 (4.0; 7.8)	52.8 (31.7; 68.7)
Ascaris	Prevalence (%) (95 CI)			
	Any intensity	31.3 (27.7; 35.2)	13.3 (10.7; 16.2)	57.5 (46.8; 66.7)
	Low intensity	20.2 (17.0; 23.3)	8.0 (5.8; 10.3)	60.4 (46.5; 71.7)
	M-H intensity	11.2 (8.8; 13.8)	5.3 (3.7; 7.2)	52.7 (29.8; 69.2)
	Mean FEC (EPG) (95 CI)	2,919 (1,999; 4,115)	1,514 (894; 2,227)	48.1 (11.0; 71.9)
Trichuris	Prevalence (%) (95 CI)			
	Any intensity	39.5 (35.7; 43.3)	18.2 (15.2; 21.2)	53.9 (44.7; 62.4)
	Low intensity	37.0 (33.2; 41.0)	16.9 (14.0;19.7)	54.3 (44.8; 63.3)
	M-H intensity	2.5 (1.3; 3.8)	1.3 (0.5; 2.3)	48.0 (-25.0; 81.0)
	Mean FEC (EPG) (95 CI)	165 (117; 222)	54 (37; 73)	67.3 (49.1; 79.8)
Hookworm	Prevalence (%) (95 CI)			
	Any intensity	10.3 (8.0; 12.8)	5.0 (3.3; 6.8)	51.5 (28.8; 69.1)
	Low intensity	10.3 (8.0; 12.8)	5.0 (3.3; 6.8)	51.5 (28.8; 69.1)
	M-H intensity	0.0 (0.0; 0.0)	0.0 (0.0; 0.0)	-
	Mean FEC (EPG) (95 CI)	23 (15; 33)	10 (6; 15)	56.5 (25.3; 78.5)
Serology	· · ·			
AsHb ELISA	Prevalence (%) (95% CI)	3.7 (2.3; 5.3)	2.2 (1.2; 3.3)	40.5 (-13.3; 71.9)
	Mean ODr	0.052 (0.040; 0.067)	0.029 (0.022; 0.038)	44.2 (19.5; 61.8)
AsLungL3 ELISA	Prevalence	54.2 (50.2; 58.2)	39.3 (35.5; 43.2)	27.5 (18.3; 35.8)
	Mean ODr	0.616 (0.537; 0.699)	0.380 (0.317; 0.448)	38.3 (24.0; 50.9)

https://doi.org/10.1371/journal.pntd.0008037.t001

Establishment of diagnostic cutoffs of serological assays. We first evaluated the IgG4 response to AsHb and AsLungL3 in a set of 500 healthy adult individuals from a non-endemic population (Belgium). This was done to establish a diagnostic cutoff for both serological assays and to record the serological reactivity that can be expected of a non-endemic adult population

(Fig 2A). The cutoffs for both tests were determined by adding 3 times the standard deviation to the mean ODr obtained by each test. For the AsHb ELISA, this resulted in a cutoff ODr of 0.36 while the cut-off ODr for the AsLungL3 ELISA positivity was set at 0.08. The results of both ELISAs were significantly correlated (Pearson's coefficient = 0.78; p < 0.0001) (Fig 2B).

Remarkably, a number of Belgian individuals showed significant reactivity to both antigens. To investigate if the reactivity to *Ascaris* antigens in these individuals was caused by exposure to *Toxocara* species (a zoonotic helminth species that is prevalent in dogs and cats in Belgium [31, 32]), their response to *Toxocara* larval excretory/secretory antigens was analyzed by a commercial ELISA. For this, we selected 65 individuals, including 35 individuals who showed elevated IgG4 responses to AsHb (ODr >0.10) and AsLungL3 (ODr >0.00) and 30 randomly selected individuals who did not show any response to either *Ascaris* antigen (ODr <0.00). Of these 65 examined Belgian serum samples, 5 were positive on the *Toxocara* ELISA (Fig 2C and 2D). Only two out of 13 and one out of eight samples that were positive for the AsHb or AsLungL3 ELISA respectively had a positive signal on the *Toxocara* ELISA. Apart from three samples with an equivocal result, all remaining samples tested negative on the *Toxocara* ELISA.

Anti-Ascaris IgG4 profiles of endemic population prior to MDA. Seroprevalence varied highly depending on the test used, with 66.2% of the population testing positive on the AsLungL3 ELISA and 5.0% testing positive on the AsHb ELISA. Compared to the copromicroscopy, two important differences were observed. First, in contrast to the age specific infection profile for copromicroscopy, for which *Ascaris* prevalence decreased as a function of increasing age, there was an increase in seroprevalence as a function of increasing age for both ELISAs (Fig 3). Seroprevalence measured by AsHb ELISA increased from 3.3% in the youngest age group to 6.7% in the oldest age group. Using the AsLungL3 ELISA, seroprevalence in the youngest age group was 45.0% and also reached a maximum in the adults older than 50. In this age group, nearly all individuals had seroconverted (89.1%). Second, although a difference in coproprevalence was observed between males (27.2%) and females (18.9%), this difference was less pronounced when applying serology (AsHb: 5.6% vs. 4.5%; AsLungL3: 66.5% vs. 65.9%). S2 Table provides the seroprevalence and mean IgG4 response to *Ascaris* antigens, based on the two ELISAs across the five age groups, both sexes and nine *kebeles*.

The outcomes of the generalized mixed models confirmed the significant differences in the test result between the three tests (copromicroscopy, AsHb and AsLungL3 EILSA) and the impact of age, but not sex or *kebele*. At an age of 18 (reference age = median age) and compared to copromicroscopy (reference test), the odds of testing positive on ELISA is 6.0 times higher (95CI: 4.80–7.48) using the AsLungL3 antigen and 6.86 times lower (95CI: 4.93–9.54) using the AsHb ELISA. The odds for a positive copro-microscopic test result decreased significantly with increasing age (OR: 0.94; 95CI: 0.92–0.96). For both ELISAs, this change in odds in a positive test result over age was significantly different. Compared to copromicroscopy, there was a net increase in odds ratio with a factor 1.08 (95CI: 1.06–1.10) and 1.12 (95CI: 1.10–1.14) per 1-year increase from the reference age for AsHb and AsLungL3 respectively. No significant differences in test results across both sexes and *kebeles* were observed.

Impact of 3 years of bi-annual MDA on anti-Ascaris IgG4 antibody levels in SAC. In the year 2015, just prior to the start of MDA, 3.7% and 54.2% of the SAC in Jimma had tested positive on the AsHb and AsLungL3 ELISA, respectively. In 2018, after three years of bi-annual deworming we observed a seroprevalence of 2.2% on the AsHb ELISA, corresponding to a non-significant (the 95 CI includes zero) reduction in seroprevalence of 40.5% (95% CI: -13.3–71.9). The seroprevalence measured by the AsLungL3 ELISA dropped to 39.3% resulting in a significant reduction of 27.5% (95% CI: 18.3; 35.8) (Table 1).



Fig 2. Serological screening of a non-endemic adult population from Belgium. Panel A shows the result of a serological screening of 500 healthy Belgian adults using the *Ascaris suum* haemoglobin (AsHb) and lung L3 extract (AsLungL3) ELISA. Horizontal dotted lines in red represent the calculated diagnostic cutoffs for each ELISA test (mean optical density ratio (ODr) of all samples + 3x the standard deviation). Group means and standard deviations are indicated with a red dotted line. Panel B shows the correlation between the IgG4 responses to AsHb and AsLungL3 in these samples. The Pearson correlation coefficient (r) and its respective significance level (*p*) is reported in the top right of the graph. Panel C and D illustrate the results of a commercial *Toxocara* ELISA and the AsHb or AsLungL3 ELISA on a subset of 65 Belgian adult sera. Here, the horizontal dotted lines represent the cutoffs that were applied to categorize the samples into negative (ratio <0.9), equivocal (0.9 ≤ ratio ≤1.1) or positive (ratio >1.1). The vertical dotted lines represent the test-specific cutoffs determined for the ELISAs with *Ascaris* antigens.

https://doi.org/10.1371/journal.pntd.0008037.g002

In 2015, the mean ODr measured in the 600 SAC by the AsHb and AsLungL3 ELISA, was 0.052 and 0.616 respectively. In 2018, this had reduced significantly to a mean ODr of 0.029 for the AsHb ELISA (reduction of 44.2%; 95% CI: 19.5–61.8) and an ODr of 0.380 for the AsLungL3 ELISA (reduction of 38.3%; 95% CI: 24.0–50.9) (Fig 4).

Agreement between FECs and ODr

Using the complete dataset (surveys 1 to 3, n = 1,800), we investigated the relationship between the IgG4 antibody responses to the *Ascaris* antigens (ODr) and the presence and intensity of STH infections (FECs) (**Fig 5**). A large number of the individuals that excreted *Ascaris* eggs were seronegative (AsHb: 96.8% (306/316) and AsLungL3: 39.2% (124/316)). In contrast, many individuals who did not excrete any *Ascaris* eggs were seropositive (AsHb: 4.2% (63/ 1,484) and AsLungL3: 56.5% (838/1,484)). Individuals excreting *Ascaris* eggs did not show a higher antibody response to either *Ascaris* antigen (t-test: AsHb: p = 0.075 and AsLungL3:



Fig 3. IgG4 responses to AsHb and AsLungL3 by age. The serological IgG4 response to *Ascaris suum* haemoglobin (AsHb) (**panel A**) and lung L3 extract (AsLungL3) (**panel B**) in 500 non-endemic Belgian adults (controls) and 1,200 individuals (600 school-aged children from survey 1 and 600 adults from survey 2) from Jimma Town (Ethiopia). Individuals from the endemic area are categorized by age. Group means and standard deviations are indicated with a full red line. Horizontal red dotted lines indicate the test-specific cutoffs. The percentage of individuals crossing the test cutoff is mentioned on top of each age group.

https://doi.org/10.1371/journal.pntd.0008037.g003

p = 0.083), nor did we observe a noteworthy relationship between the *Ascaris* FECs measured by Kato-Katz and the intensity of the IgG4 response to AsHb (Pearson's coefficient = 0.03 p = 0.223) or AsLungL3 (Pearson's coefficient = 0.05, p = 0.04).



Fig 4. IgG4 responses to AsHb and AsLungL3 before and three years into the deworming program. The IgG4 responses to *Acaris suum* haemoglobin (AsHb) (A) and lung L3 extract (AsLungL3) (B) measured in 600 school-aged children from Jimma, Ethiopia, before (2015) and after (2018) 3 years of biannual mass drug administration with mebendazole. Horizontal red dotted lines represent the calculated diagnostic cutoffs for each ELISA test. Group averages and their standard deviation is indicated by full red lines.

https://doi.org/10.1371/journal.pntd.0008037.g004



Fig 5. Relationship between *Ascaris* **fecal egg counts and antibody response to** *Ascaris* **antigens.** IgG4 antibodies to *Ascaris suum* haemoglobin (AsHb) (A) and lung L3 extract (AsLungL3) (B) measured in the 1,800 sampled SAC and adult individuals from Jimma Town, Ethiopia are plotted against fecal egg counts (FECs) obtained by Kato-Katz. FECs are expressed as eggs per gram of stool (EPG)). The horizontal red dotted lines represent the assay-specific cutoffs.

https://doi.org/10.1371/journal.pntd.0008037.g005

Assessment of cross-reactivity

Using the complete dataset again (including all 1,800 samples), we investigated the effect of infection with other STHs on the serological IgG4 response to the *Ascaris* antigens (cross-reac-tivity). For this, individuals were grouped according to the STHs detected in their stool sample.



Fig 6. Impact of helminth coinfections on serological response to *A. suum* **antigens.** These plots show the IgG4 antibody reactivity to *A. suum* haemoglobin (AsHb) (A) and total lungL3 extract (AsLungL3) (B) in 1,200 SAC and 600 adults from Jimma Town, Ethiopia. Individuals are categorized by soil-transmitted helminth (STH) status. Group means and standard deviations are indicated in red (No = no STH eggs detected; AL: only positive for *Ascaris*; TT: only positive for *Trichuris*; HW: only positive for hookworms; TT&HW: both *Trichuris* and hookworm eggs detected in the stool; AL and TT and/or HW: *Ascaris* eggs detected by copromicroscopy together with *Trichuris* and /or hookworm eggs).

https://doi.org/10.1371/journal.pntd.0008037.g006

Depending on the type of eggs detected in the stool during Kato-Katz examination, individuals were labeled as not infected, infected with *Ascaris*, *Trichuris* or hookworm only, having a mixed infection with *Trichuris* and hookworm or being infected with *Ascaris* plus at least one additional STH species. The serological responses of these individuals to AsHb and AsLungL3 is presented in **Fig 6**. Both graphs indicate that, independent of the presence or type of infection present in the individual at the time of sampling, average antibody responses to the *Ascaris* antigens are very similar.

Discussion

The scientific community has recently summarized the desired characteristics for diagnostic tools across different phases of MDA programs. As substantiated by other NTD programs, serology holds promise as an alternative diagnostic tool when the program approaches the endgame [7, 33]. To supplement the scarcity of data on baseline serological profiles in human populations, we measured anti-*Ascaris* IgG4 antibodies in SAC and adults residing in both an STH endemic (Jimma Town, Ethiopia) and a non-endemic (Belgium) area using two different in-house ELISAs based on two *A. suum* antigen preparations. In addition, we compared the infection profiles in SAC before and after 3 years of biannual school-based MDA using copromicroscopy and serology.

Adult population remains a reservoir for STH transmission

Although both coproprevalence and mean FECs dropped significantly as a function of increasing age, the adult population in Jimma Town still hosts a substantial reservoir of adult worms, capable of maintaining transmission in the population. Both deterministic and stochastic STH transmission models [5], as well as recent findings by Dunn et al. [34], who found that adults were major contributors to total STH prevalence and FECs, seem to underpin the notion that broadening coverage to encompass all age classes will likely be a more effective approach for the long-term control of STHs than restricting deworming to SAC only. However, at this moment, the impact of school-based MDA on the STH transmission in the adult population is not well known. More longitudinal monitoring and evaluation studies are needed to evaluate the long-term impact of these control strategies across the whole population.

Increased antibody response against *Ascaris* antigens in some Belgian adults

In this study, the cutoff values for both ELISAs (AsHb and AsLungL3 ELISA) were determined using 500 serum samples from healthy adult blood donors in Belgium as negative population. Despite the absence of STH in Belgium, a number of serum samples did show increased antibody responses to AsHb and AsLungL3 antigens. There are a number of possible explanations for these findings, including zoonotic infections with canine or feline *Toxocara* spp. or the porcine helminth *Ascaris suum. Toxocara* spp. are closely related to *A. lumbricoides* and are prevalent in domesticated cats and dogs in Belgium [31, 32]. To investigate the presence of *Toxocara* infections, a set of 35 samples that showed elevated IgG4 responses to the *Ascaris* antigens and 30 samples that did not respond to either *Ascaris* antigens were screened on a commercial *Toxocara* ELISA. Only two *Ascaris* seropositive samples showed a positive result on the *Toxocara* ELISA, suggesting that there is limited cross reactivity between the respective ELISAs. The porcine roundworm *A. suum* is still highly prevalent in the Belgian pig industry [14, 16] and has been shown to infect humans [35–38]. We also cannot exclude that exposure to STHs or *Toxocara* species while travelling abroad could have occurred [39, 40], as we did not have any background information on our Belgian serum donors. These elevated antibody

levels against *Ascaris* antigens in a number of Belgian samples does however confirm a serological study performed in the Netherlands that found 45 out of 629 (7.2%) children to be *Ascaris*seropositive [41]. Yet, here too, it remains unknown what the source of infections is, and to what extent these infections have any health consequences. For future serological studies, the availability of sera from young, unexposed, non-endemic individuals would be of value for assay cut-off determination.

Exposure to the infectious stages of Ascaris reaches beyond SAC

Anti-AsLungL3 seroprevalence was 45.0% in the youngest age group and steadily increased with age reaching nearly 90% seroprevalence in individuals over the age of 50. Such findings can be interpreted as evidence that individuals are continuously exposed to the larval stages of *Ascaris* from a very early age onwards, and that the absence of *Ascaris* eggs in the stool of older individuals does not necessarily imply the absence of exposure to larval stages. The fact that *Ascaris* exposure is of all ages is fostered by the fact that patent infections were detected in all age groups. The oldest individual found to excrete *A. lumbricoides* eggs was 85 years old.

The AsLungL3 extract contains promising antigens for further diagnostic tool development

Serological analysis of the sera collected in this study revealed a significant difference in IgG4 antibody reactivity to the two Ascaris antigen preparations. Relatively few individuals had an elevated IgG4 response to AsHb, while many individuals showed elevated sero-reactivity to the AsLungL3 antigen. One reason for the low number of AsHb seropositive individuals could be due to the way the assay cut-off was determined. The cut-off for the AsHb assay was notably higher than the cut-off for the AsLungL3 ELISA. This is the direct consequence of the fact that more Belgian individuals had an elevated antibody response to the AsHb than to the AsLungL3 extract (Fig 2B), resulting in a significantly larger standard deviation around the average ODr, and thus a higher cut-off for AsHb assay (AsHb: 0.36 ODr vs. AsLungL3: 0.08 ODr). However, this fact alone clearly does not explain the large discrepancy between the results obtained by both ELISAs. Overall, there appears to be a much more pronounced IgG4 antibody response to the AsLungL3 antigen compared to the AsHb antigen. This could partly be described to the fact that the AsLungL3 antigen is a complete larval extract that contains a multitude of larval antigens and epitopes for antibodies to bind to, whereas, AsHb is a single purified native antigen with a limited number of antibody binding targets. These results are however in correspondence with the results of a recent evaluation of the AsLungL3 extract as diagnostic antigen to measure exposure in piglets experimentally infected with A. suum. This study also found that the antibody response to AsLungL3 developed faster and was more pronounced than the one directed to AsHb [13]. Hence, determining the antigenic constitution of the AsLungL3 extract is of particular interest for future diagnostic test development.

By grouping individuals according to the type of patent infection they carried (as verified by copromicroscopy), we wished to evaluate the presence of cross-reactive antibodies to the *A. suum* antigens. However, antibody responses to these antigens were very similar across all different groups and it was not possible to draw conclusions with regards to the species-specificity of the assays. Based on qPCR results from a recent survey in the Jimma area, we do however know that the prevalence of other parasitic species that might cause antibody cross-reactivity such as *Schistosoma mansoni*, *Strongyloides stercoralis* and *Taenia* spp in SAC is very low (3.5%, 3.5% and 2.7% respectively). Moreover, Jimma town is considered hypo-endemic for *Onchocerca volvulus* [42] and non-endemic for other filarial infections including *Mansonella perstans* [43], *Loa loa* [44] or *Wuchereria bancrofti* [45, 46].

Three years of biannual MDA reduces STH prevalence and serological response to *Ascaris* antigens in SAC

This study also compared coprological and serological results collected from SAC at the start of the STH control program in 2015 with those collected in 2018, three years into the program. The effects of the control program were clearly visible. Overall STH infection had dropped from 57.7% to 28.0% (51.5% reduction), and so did the intensity of the infections (Ascaris: 48.2%, Trichuris: 67.3% and hookworm: 69.6% reduction). Interestingly, the control program also led to a significantly reduced IgG4 response to both AsHb and AsLungL3, resulting in less seropositive individuals (AsHb: 40.9% reduction; AsLungL3: 27.4% reduction) and significantly lower mean ODr levels measured in the SAC (AsHb; 44.2% reduction; AsLungL3: 38.2% reduction). These findings support the results presented in the study of Vlaminck and colleagues from 2016 [19] where anti-AsHb IgG4 levels were evaluated in an Indonesian community subjected to community wide MDA with albendazole for lymphatic filariasis and soiltransmitted helminthiasis. Even though, in that particular population, significantly higher number of AsHb seropositive samples were detected at baseline (67.7% in 2002), the seroprevalence significantly reduced during the course of the intervention (22.6% in 2004, 23.0% in 2006 and 14.3% in 2009). At the same time, average A. lumbricoides coproprevalence in the community reduced with 22.8% (from 38.2% in 2002 to 29.5% in 2009) while community egg output reduced with 81.6% (from 698 EPG in 2002 to 128 EPG in 2009).

Absence of association between coprological and serological results

It is important to note that many of the examined individuals excreting *Ascaris* eggs were seronegative (96.2% using the AsHb and 43.6% using the AsLungL3 assay) and that 82.1% (AsHb) and 79.9% (AsLungL3) of seropositive individuals did not excrete any *Ascaris* eggs. Other studies that compared anti-*Ascaris* antibody responses to adult worm counts and FECs also reported the absence of a significant association between antibody titers and worm load or egg production [19, 47, 48]. Moreover, King *et al.* [47] also noted a negative association between total IgG responses to larval *A. lumbricoides* antigen and infection intensity in young children (5–11 years old). There is no real straightforward explanation for this seemingly inversed correlation. One possibility is that continued *Ascaris* exposure acts as an ongoing booster to maintain elevated protective immune responses, hereby preventing the development of new patent infections. This is supported by earlier reports that have shown that Th2-mediated responses play an important role in the age-dependent reduction of intestinal helminth infections in humans [49, 50].

Conclusions

During *Ascaris* infection, antibodies are raised to immunogenic components produced by the worm and which are detected and processed by the immune system. Such antibodies are unique in their capacity to serve as a biomarker to identify individuals that have experienced exposure to this infection. Measuring infection-induced antibodies has proven an objective way to monitor a population's exposure to multiple NTDs [12, 51–54]. This study clearly demonstrated the positive impact of the MDA program on STH infection prevalence in Jimma Town, Ethiopia and further underpins the idea that it is possible to evaluate the impact of control efforts by measuring the changes in antibody responses to STH antigens. Combining a number of antibody assays in a multiplex format would provide a great opportunity to integrate monitoring and surveillance of a number of infectious diseases [55]. In order to progress towards a species-specific antibody test for *Ascaris*, it would be of special interest to identify

the specific immunodominant antigens present in the AsLungL3 extract. In addition, it is absolutely necessary to collect more data on how to interpret and apply serological results. We need a far better understanding of the serological profiles produced by populations with different backgrounds of exposure to *Ascaris* and STHs in general.

Supporting information

S1 Table. The prevalence and intensity of STH infections based on copromicroscopy in *Jimma Town, 2015.* The table summarizes the prevalence and intensity of *Ascaris lumbricoides, Trichuris trichiura* and hookworm infections based on copromicroscopy (Kato-Katz, Mini-FLOTAC and McMaster) across different age groups, both sexes and nine *kebeles* at the start of the national deworming program. The intensity of infection is measured by mean fecal egg counts (FECS; expressed in eggs per gram of stool (EPG)). (DOCX)

S2 Table. The prevalence and intensity of STH infections based on serology in Jimma Town, 2015. This table summarizes the prevalence and intensity of *Ascaris lumbricoides* infections based on serology across five different age groups, both sexes and nine *kebeles* at the start of the national deworming program in Jimma Town, Ethiopia. The intensity of infection is measured by mean optical density ratio (ODr). (DOCX)

S1 Info. The complete study dataset. (CSV)

Acknowledgments

We would like to thank the STH laboratory staff at Jimma University who helped performing the data collection and laboratory examination. We are also grateful to the staff of the Laboratory for Parasitology at the Faculty of Veterinary medicine of Ghent University for their unreserved support during performance of the serological diagnosis.

Author Contributions

Conceptualization: Johnny Vlaminck, Zeleke Mekonnen, Peter Geldhof, Bruno Levecke.

Data curation: Daniel Dana, Johnny Vlaminck.

Formal analysis: Daniel Dana, Johnny Vlaminck.

Funding acquisition: Zeleke Mekonnen, Peter Geldhof, Bruno Levecke.

Investigation: Daniel Dana, Johnny Vlaminck, Mio Ayana, Bamlaku Tadege.

Methodology: Daniel Dana, Johnny Vlaminck, Mio Ayana, Bamlaku Tadege.

Project administration: Johnny Vlaminck, Bruno Levecke.

Resources: Zeleke Mekonnen, Bruno Levecke.

Supervision: Johnny Vlaminck, Zeleke Mekonnen, Peter Geldhof, Bruno Levecke.

Validation: Johnny Vlaminck.

Visualization: Johnny Vlaminck.

Writing - original draft: Daniel Dana, Johnny Vlaminck.

Writing – review & editing: Johnny Vlaminck, Zeleke Mekonnen, Peter Geldhof, Bruno Levecke.

References

- GBD 2017 DALYs and HALE Collaborators. Global, regional, and national disability-adjusted life-years (DALYs) for 359 diseases and injuries and healthy life expectancy (HALE) for 195 countries and territories, 1990–2017: a systematic analysis for the Global Burden of Disease Study 2017. Lancet. 2018; 392:1859–922. https://doi.org/10.1016/S0140-6736(18)32335-3 PMID: 30415748
- GBD 2015 Disease and Injury Incidence and Prevalence Collaborators. Global, regional, and national incidence, prevalence, and years lived with disability for 310 diseases and injuries, 1990–2015: a systematic analysis for the Global Burden of Disease Study 2015. Lancet. 2016; 388(10053):1545–602. https://doi.org/10.1016/S0140-6736(16)31678-6 PMID: 27733282
- 3. World Health Organization. Crossing the billion. Lymphatic filariasis, onchocerciasis, schistosomiasis, soil-transmitted helminthiasis and trachoma: preventive chemotherapy for neglected tropical diseases. Geneva, Switzerland, 2017.
- Asbjornsdottir KH, Ajjampur SSR, Anderson RM, Bailey R, Gardiner I, Halliday KE, et al. Assessing the feasibility of interrupting the transmission of soil-transmitted helminths through mass drug administration: The DeWorm3 cluster randomized trial protocol. PLoS Negl Trop Dis. 2018; 12(1):e0006166. https://doi.org/10.1371/journal.pntd.0006166 PMID: 29346377
- Anderson RM, Turner HC, Truscott JE, Hollingsworth TD, Brooker SJ. Should the goal for the treatment of soil-transmitted helminth (STH) infections be changed from morbidity control in children to community-wide transmission elimination? PLoS Negl Trop Dis. 2015; 9(8):e0003897. <u>https://doi.org/10.1371/journal.pntd.0003897</u> PMID: 26291538
- 6. World Health Organization. Soil-transmitted helminthiases: eliminating soil-transmitted helminthiases as a public healt problem in children: progress report 2001–2010 and strategic plan 2011–2020. Geneva, Switserland, 2012.
- 7. World Health Organization. Onchocerciasis: guidelines for stopping mass drug administration and verifying elimination of human onchocerciasis. Geneva, Switzerland, 2016.
- World Health Organization. Schistosomiasis: Progress report 2001–2011, strategic plan 2012–2020. Geneva, Switzerland, 2013.
- 9. World Health Organization. Basic laboratory methods in medical parasitology. Geneva, Switzerland, 1991.
- Lim MD, Brooker SJ, Belizario VY Jr, Gay-Andrieu F, Gilleard J, Levecke B, et al. Diagnostic tools for soil-transmitted helminths control and elimination programs: A pathway for diagnostic product development. PLoS Negl Trop Dis. 2018; 12(3):e0006213. <u>https://doi.org/10.1371/journal.pntd.0006213</u> PMID: 29494581
- Cantey PT, Roy SL, Boakye D, Mwingira U, Ottesen EA, Hopkins AD, et al. Transitioning from river blindness control to elimination: steps toward stopping treatment. Int Health. 2018; 10(suppl_1):i7–i13. https://doi.org/10.1093/inthealth/ihx049 PMID: 29471338
- 12. Won KY, Sambou S, Barry A, Robinson K, Jaye M, Sanneh B, et al. Use of Antibody Tools to Provide Serologic Evidence of Elimination of Lymphatic Filariasis in The Gambia. Am J Trop Med Hyg. 2017.
- Vandekerckhove E, Vlaminck J, Geldhof P. Evaluation of serology to measure exposure of piglets to Ascaris suum during the nursery phase. Vet Parasitol. 2017; 246:82–7. <u>https://doi.org/10.1016/j.vetpar.</u> 2017.09.008 PMID: 28969785
- Vlaminck J, Nejsum P, Vangroenweghe F, Thamsborg SM, Vercruysse J, Geldhof P. Evaluation of a serodiagnostic test using *Ascaris suum* haemoglobin for the detection of roundworm infections in pig populations. Veterinary Parasitology. 2012; 189(2–4):267–73. <u>https://doi.org/10.1016/j.vetpar.2012.04</u>. 024 PMID: 22560331
- **15.** Martinez-Perez JM, Vandekerckhove E, Vlaminck J, Geldhof P, Martinez-Valladares M. Serological detection of *Ascaris suum* at fattening pig farms is linked with performance and management indices. Vet Parasitol. 2017; 248:33–8. https://doi.org/10.1016/j.vetpar.2017.10.009 PMID: 29173538
- Vlaminck J, Dusseldorf S, Heres L, Geldhof P. Serological examination of fattening pigs reveals associations between *Ascaris suum*, lung pathogens and technical performance parameters. Vet Parasitol. 2015; 210(3–4):151–8. https://doi.org/10.1016/j.vetpar.2015.04.012 PMID: 25952722
- Vandekerckhove E, Vlaminck J, Sacristan RDP, Geldhof P. Effect of strategic deworming on Ascaris suum exposure and technical performance parameters in fattening pigs. Vet Parasitol. 2019; 268:67– 72. https://doi.org/10.1016/j.vetpar.2019.03.006 PMID: 30981308

- Lassen B, Geldhof P, Halli O, Vlaminck J, Oliviero C, Orro T, et al. Anti-Ascaris suum IgG antibodies in fattening pigs with different respiratory conditions. Vet Parasitol. 2019; 265:85–90. https://doi.org/10. 1016/j.vetpar.2018.12.005 PMID: 30638525
- Vlaminck J, Supali T, Geldhof P, Hokke CH, Fischer PU, Weil GJ. Community Rates of IgG4 Antibodies to Ascaris Haemoglobin Reflect Changes in Community Egg Loads Following Mass Drug Administration. PLoS Negl Trop Dis. 2016; 10(3):e0004532. <u>https://doi.org/10.1371/journal.pntd.0004532</u> PMID: 26991326
- Levecke B, Montresor A, Albonico M, Ame SM, Behnke JM, Bethony JM, et al. Assessment of anthelmintic efficacy of mebendazole in school children in six countries where soil-transmitted helminths are endemic. PLoS Negl Trop Dis. 2014; 8(10):e3204. https://doi.org/10.1371/journal.pntd.0003204 PMID: 25299391
- Mekonnen Z, Meka S, Ayana M, Bogers J, Vercruysse J, Levecke B. Comparison of individual and pooled stool samples for the assessment of soil-transmitted helminth infection intensity and drug efficacy. PLoS Negl Trop Dis. 2013; 7(5):e2189. <u>https://doi.org/10.1371/journal.pntd.0002189</u> PMID: 23696905
- Vercruysse J, Behnke JM, Albonico M, Ame SM, Angebault C, Bethony JM, et al. Assessment of the anthelmintic efficacy of albendazole in school children in seven countries where soil-transmitted helminths are endemic. PLoS Negl Trop Dis. 2011; 5(3):e948. <u>https://doi.org/10.1371/journal.pntd.</u> 0000948 PMID: 21468309
- Barda B, Cajal P., Villagran E., Cimino R., Juarez M., Krolewiecki A., Rinaldi L., Cringoli G., Burioni R. and Albonico M. Mini-FLOTAC, Kato-Katz and McMaster: three methods, one goal; highlights from north Argentina. Parasites & vectors. 2014; 7(1).
- 24. Starworms. Visual tutorials 2019 [cited 2019 November 26, 2019]. Available from: <u>https://www.</u>starworms.org/tools/overview/visual-tutorials.
- Levecke B, Behnke JM, Ajjampur SS, Albonico M, Ame SM, Charlier J, et al. A comparison of the sensitivity and fecal egg counts of the McMaster egg counting and Kato-Katz thick smear methods for soiltransmitted helminths. PLoS Negl Trop Dis. 2011; 5(6):e1201. https://doi.org/10.1371/journal.pntd. 0001201 PMID: 21695104
- Cringoli G RL, Maurelli MP, Utzinger J. FLOTAC: new multivalent techniques for qualitative and quantitative copromicroscopic diagnosis of parasites in animals and humans. Nat Protoc. 2010; 5:503–15. https://doi.org/10.1038/nprot.2009.235 PMID: 20203667
- Cringoli G RL, Levecke B, Bosco A, Vercruysse J, Utzinger J, Rinaldi L. The Mini-FLOTAC technique for the diagnosis of helminth and protozoan infections in humans and animals. nature protocols. 2017; 12 (9):1723–32. https://doi.org/10.1038/nprot.2017.067 PMID: 28771238
- Santra A, Bhattacharya T, Chowdhury A, Ghosh A, Ghosh N, Chatterjee BP, et al. Serodiagnosis of ascariasis with specific IgG4 antibody and its use in an epidemiological study. Trans R Soc Trop Med Hyg. 2001; 95(3):289–92. https://doi.org/10.1016/s0035-9203(01)90236-6 PMID: 11490999
- 29. Chatterjee BP, Santra A, Karmakar PR, Mazumder DN. Evaluation of IgG4 response in ascariasis by ELISA for serodiagnosis. Trop Med Int Health. 1996; 1(5):633–9. <u>https://doi.org/10.1111/j.1365-3156.1996.tb00088.x PMID: 8911447</u>
- **30.** World Health Organization. Helminth control in school-age children: a guide for managers of control programmes. Geneva, Switserland: World Healt Organisation, 2011.
- Claerebout E, Casaert S, Dalemans AC, De Wilde N, Levecke B, Vercruysse J, et al. Giardia and other intestinal parasites in different dog populations in Northern Belgium. Vet Parasitol. 2009; 161(1–2):41– 6. https://doi.org/10.1016/j.vetpar.2008.11.024 PMID: 19155136
- Ma G, Holland CV, Wang T, Hofmann A, Fan CK, Maizels RM, et al. Human toxocariasis. Lancet Infect Dis. 2018; 18(1):e14–e24. Epub 2017/08/07. https://doi.org/10.1016/S1473-3099(17)30331-6 PMID: 28781085
- **33.** World Health Organization. Lymphatic filariasis: a manual for national elimination programmes. Geneva, Switserland.
- 34. Dunn JC, Bettis AA, Wyine NY, Lwin AMM, Lwin ST, Su KK, et al. A cross-sectional survey of soil-transmitted helminthiases in two Myanmar villages receiving mass drug administration: epidemiology of infection with a focus on adults. Parasit Vectors. 2017;10. https://doi.org/10.1186/s13071-016-1942-2
- 35. Bendall RP, Barlow M, Betson M, Stothard JR, Nejsum P. Zoonotic ascariasis, United Kingdom. Emerg Infect Dis. 2011; 17(10):1964–6. https://doi.org/10.3201/eid1710.101826 PMID: 22000387
- Nejsum P, Bertelsen MF, Betson M, Stothard JR, Murrell KD. Molecular evidence for sustained transmission of zoonotic Ascaris suum among zoo chimpanzees (Pan troglodytes). Vet Parasitol. 2010; 171 (3–4):273–6. https://doi.org/10.1016/j.vetpar.2010.03.030 PMID: 20400228

- Schneider R, Auer H. Incidence of Ascaris suum-specific antibodies in Austrian patients with suspected larva migrans visceralis (VLM) syndrome. Parasitol Res. 2016; 115(3):1213–9. Epub 2015/12/08. https://doi.org/10.1007/s00436-015-4857-5 PMID: 26637313
- Avery RH, Wall LA, Verhoeve VI, Gipson KS, Malone JB. Molecular Confirmation of Ascaris suum: Further Investigation into the Zoonotic Origin of Infection in an 8-Year-Old Boy with Loeffler Syndrome. Vector Borne Zoonotic Dis. 2018; 18(11):638–40. <u>https://doi.org/10.1089/vbz.2018.2306</u> PMID: 30085905
- 39. Van Den Broucke S, Kanobana K, Polman K, Soentjens P, Vekemans M, Theunissen C, et al. Toxocariasis diagnosed in international travelers at the Institute of Tropical Medicine, Antwerp, Belgium, from 2000 to 2013. PLoS Negl Trop Dis. 2015; 9(3):e0003559. https://doi.org/10.1371/journal.pntd.0003559 PMID: 25746418
- Pimenta D, Dawood N. An unusual cause of pancreatitis in a 46-year-old returning traveller. BMJ Case Rep. 2015;2015. Epub 2015/08/05.
- 41. Pinelli E, Willers SM, Hoek D, Smit HA, Kortbeek LM, Hoekstra M, et al. Prevalence of antibodies against Ascaris suum and its association with allergic manifestations in 4-year-old children in The Netherlands: the PIAMA birth cohort study. Eur J Clin Microbiol Infect Dis. 2009; 28(11):1327–34. https://doi.org/10.1007/s10096-009-0785-6 PMID: 19644714
- 42. Dana D, Debalke S, Mekonnen Z, Kassahun W, Suleman S, Getahun K, et al. A community-based cross-sectional study of the epidemiology of onchocerciasis in unmapped villages for community directed treatment with ivermectin in Jimma Zone, southwestern Ethiopia. BMC Public Health. 2015; 15:595. Epub 2015/07/02. https://doi.org/10.1186/s12889-015-1888-x PMID: 26130117
- 43. Simonsen PE, Onapa AW, Asio SM. *Mansonella perstans* filariasis in Africa. Acta Trop. 2011; 120 Suppl 1:S109–20.
- Zoure HG, Wanji S, Noma M, Amazigo UV, Diggle PJ, Tekle AH, et al. The geographic distribution of Loa loa in Africa: results of large-scale implementation of the Rapid Assessment Procedure for Loiasis (RAPLOA). PLoS Negl Trop Dis. 2011; 5(6):e1210. <u>https://doi.org/10.1371/journal.pntd.0001210</u> PMID: 21738809
- 45. Shiferaw W, Kebede T, Graves PM, Golasa L, Gebre T, Mosher AW, et al. Lymphatic filariasis in western Ethiopia with special emphasis on prevalence of *Wuchereria bancrofti* antigenaemia in and around onchocerciasis endemic areas. Trans R Soc Trop Med Hyg. 2012; 106(2):117–27. https://doi.org/10. 1016/j.trstmh.2011.10.006 PMID: 22154976
- 46. Deribe K, Meribo K, Gebre T, Hailu A, Ali A, Aseffa A, et al. The burden of neglected tropical diseases in Ethiopia, and opportunities for integrated control and elimination. Parasit Vectors. 2012; 5:240. <u>https:// doi.org/10.1186/1756-3305-5-240 PMID: 23095679</u>
- King EM, Kim HT, Dang NT, Michael E, Drake L, Needham C, et al. Immuno-epidemiology of Ascaris lumbricoides infection in a high transmission community: antibody responses and their impact on current and future infection intensity. Parasite Immunol. 2005; 27(3):89–96. <u>https://doi.org/10.1111/j.1365-3024.2005.00753.x</u> PMID: 15882235
- Bhattacharyya T, Santra A, Majumder DNG, Chatterjee BP. Possible approach for serodiagnosis of ascariasis by evaluation of immunoglobulin G4 response using *Ascaris lumbricoides* somatic antigen. Journal of Clinical Microbiology. 2001; 39(8):2991–4. https://doi.org/10.1128/JCM.39.8.2991-2994. 2001 PMID: 11474031
- Jackson JA, Turner JD, Rentoul L, Faulkner H, Behnke JM, Hoyle M, et al. T helper cell type 2 responsiveness predicts future susceptibility to gastrointestinal nematodes in humans. J Infect Dis. 2004; 190 (10):1804–11. https://doi.org/10.1086/425014 PMID: 15499537
- Turner JD, Faulkner H, Kamgno J, Cormont F, Van Snick J, Else KJ, et al. Th2 cytokines are associated with reduced worm burdens in a human intestinal helminth infection. J Infect Dis. 2003; 188(11):1768– 75. https://doi.org/10.1086/379370 PMID: 14639550
- Feeser KR, Cama V, Priest JW, Thiele EA, Wiegand RE, Lakwo T, et al. Characterizing Reactivity to Onchocerca volvulus Antigens in Multiplex Bead Assays. Am J Trop Med Hyg. 2017; 97(3):666–72. https://doi.org/10.4269/ajtmh.16-0519 PMID: 28722605
- Lammie PJ, Moss DM, Brook Goodhew E, Hamlin K, Krolewiecki A, West SK, et al. Development of a new platform for neglected tropical disease surveillance. Int J Parasitol. 2012; 42(9):797–800. <u>https:// doi.org/10.1016/j.ijpara.2012.07.002</u> PMID: 22846784
- 53. Hamlin KL, Moss DM, Priest JW, Roberts J, Kubofcik J, Gass K, et al. Longitudinal monitoring of the development of antifilarial antibodies and acquisition of *Wuchereria bancrofti* in a highly endemic area of Haiti. PLoS Negl Trop Dis. 2012; 6(12):e1941. <u>https://doi.org/10.1371/journal.pntd.0001941</u> PMID: 23236534
- Silva-Moraes V, Shollenberger LM, Castro-Borges W, Rabello ALT, Harn DA, Medeiros LCS, et al. Serological proteomic screening and evaluation of a recombinant egg antigen for the diagnosis of low-

intensity *Schistosoma mansoni* infections in endemic area in Brazil. PLoS Negl Trop Dis. 2019; 13(3): e0006974. https://doi.org/10.1371/journal.pntd.0006974 PMID: 30870412

55. Arnold BF, Scobie HM, Priest JW, Lammie PJ. Integrated Serologic Surveillance of Population Immunity and Disease Transmission. Emerg Infect Dis. 2018; 24(7):1188–94. <u>https://doi.org/10.3201/</u> eid2407.171928 PMID: 29912680